Author's response to reviews

Title: A Dirichlet Process Model for Classifying and Forecasting Epidemic Curves

Authors:

Elaine O Nsoesie (onelaine@vbi.vt.edu)
Scotland C Leman (leman@vt.edu)
Madhav V Marathe (mmarathe@vbi.vt.edu)

Version: 2
Date: 19 October 2013

Author's response to reviews:

Dear Editor:

We thank you and the reviewer for your comments and suggestions. We have made extensive revisions to the content of the manuscript. We also applied the method to data acquired from the Centers for Disease Control and Prevention (CDC) as requested by the reviewer. Thank you.

Sincerely,

Elaine Nsoesie